Author's response to reviews

Title: Seroprevalence of Immunoglobulin G antibodies against pertussis toxin among asymptomatic medical students in the west of Iran: a cross sectional study

Authors:

Seyyed Hamid - Hashemi - (mitraranjbar@yahoo.com)
Mitra - Ranjbar - (mitraranjbar@yahoo.com)
Mehrdad - Hajilooi - (mitraranjbar@yahoo.com)
Mohammad-Ali - Seif-Rabiei - (mitraranjbar@yahoo.com)
Mahnaz - Bolandi - (mitraranjbar@yahoo.com)
Javad - Moghimi - (mitraranjbar@yahoo.com)

Version: 3 Date: 19 April 2009

Author's response to reviews: see over
Dear Editor
Many thanks for your faithful comments about our manuscript. We went through all the comments and revised the manuscript accordingly, hope to satisfy the reviewers. All changes and revisions were tacked in the text. Also, point by point response to the comments was provided.
The manuscript was formatted according the instruction for authors of the journal. Also authors' contributions, competing interest and list of abbreviations were added also.
In case there is any problem no hesitate to contact us. Please attached find the revised manuscript and point by point response to the reviewers.

Regards,
M. Ranjbar, MD

1- Authors contribution was added.
2- Statement of competing interest was added.
3- Questions were answered.

The questions that are answered are:

**Financial competing interests**
- In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.
**Response:** Nothing

- Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.
**Response:** None

- Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.
**Response:** No

- Do you have any other financial competing interests? If so, please specify.

**Non-financial competing interests**

**Response:** No

Are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.
**Response:** None